\begin{thebibliography}{10}

\bibitem{bonda98}
A.~Bondavalli, I.~Mura, and I.~Majzik.
\newblock {Automated dependability analysis of {UML} designs}.
\newblock In {\em Proc. of Second IEEE International Symposium on
  Object-Oriented Real-Time Distributed Computing}, 1998.

\bibitem{mrspn}
H.~Choi, V.~Kulkarni, and K.~S. Trivedi.
\newblock ``{Markov} {Regenerative} {Stochastic} {Petri} {Net}''.
\newblock {\em Performance Evaluation}, 20(1--3):337--357, 1994.

\bibitem{Globus:97}
I.~Foster and C.~Kesselman.
\newblock Globus: A metacomputing infrastructure toolkit.
\newblock {\em International Journal of Supercomputer Applications},
  11(2):115--128, 1997.

\bibitem{Vlissides:94}
E.~Gamma, R.~Helm, R.~Johnson, and J.~Vlissides.
\newblock {\em {Design Patterns: Elements of Reusable Object-Oriented
  Software}}.
\newblock Addison-Wesley, Reading, MA, 1995.

\bibitem{Hatcliff:03}
J.~Hatcliff, W.~Deng, M.~Dwyer, G.~Jung, and V.~Prasad.
\newblock {Cadena: An Integrated Development, Analysis, and Verification
  Environment for Component-based Systems}.
\newblock In {\em Proceedings of the 25th International Conference on Software
  Engineering}, Portland, OR, May 2003.

\bibitem{spnp-new}
C.~Hirel, B.~Tuffin, and K.~S. Trivedi.
\newblock ``{SPNP}: {Stochastic} {Petri} {Nets}. {Version} 6.0''.
\newblock {\em Lecture Notes in Computer Science 1786}, 2000.

\bibitem{hoeben00}
F.~Hoeben.
\newblock ``{Using} {UML} models for performance calculation''.
\newblock In {\em Proc. of Workshop on Software and Performance}, 2000.

\bibitem{fspn-1}
G.~Horton, V.~Kulkarni, D.~Nicol, and K.~S. Trivedi.
\newblock ``{Fluid} stochastic {Petri} nets: {Theory}, application and solution
  techniques''.
\newblock {\em Journal of Operations Research}, 405, 1998.

\bibitem{avail-vax}
O.~Ibe, A.~Sathaye, R.~Howe, and K.~S. Trivedi.
\newblock ``{Stochastic} {Petri} net modeling of {VAXCluster} availability''.
\newblock In {\em Proc. of Third International Workshop on Petri Nets and
  Performance Models}, pages 142--151, Kyoto, Japan, 1989.

\bibitem{poll-perf}
O.~Ibe and K.~S. Trivedi.
\newblock ``{Stochastic} {Petri} net models of polling systems''.
\newblock {\em IEEE Journal on Selected Areas in Communications},
  8(9):1649--1657, December 1990.

\bibitem{queue-perf}
O.~Ibe and K.~S. Trivedi.
\newblock ``{Stochastic} {Petri} net analysis of finite--population queueing
  systems''.
\newblock {\em Queueing Systems: Theory and Applications}, 8(2):111--128, 1991.

\bibitem{king99}
P.~King and R.~Pooley.
\newblock ``{Derivation} of {Petri} net performance models from {UML}
  specifications of communication software''.
\newblock In {\em Proc. of XV Performance Engineering Workshop}, 1997.

\bibitem{menasce98}
D.~A. Menasce and H.~A. Gomaa.
\newblock ``{On} a language based method for software performance engineering
  of client/server systems''.
\newblock In {\em Proc. of Workshop on Software and Performance}, 1998.

\bibitem{rel-srn}
J.~Muppala, G.~Ciardo, and K.~S. Trivedi.
\newblock ``{Stochastic} {reward} nets for reliability prediction''.
\newblock {\em Communications in Reliability, Maintainability and
  Serviceability: An International Journal Published by SAE Internationa},
  1(2):9--20, July 1994.

\bibitem{peterson81}
J.~L. Peterson.
\newblock {\em Petri Net Theory and the Modeling of Systems}.
\newblock Prentice-Hall, 1981.

\bibitem{petriu00}
D.~Petriu, C.~Shousha, and A.~Jalnapurkar.
\newblock ``{Architecture}--based performance analysis applied to a
  telecommunication system''.
\newblock {\em IEEE Trans. on Software Engineering}, November 2000.

\bibitem{pooley99}
R.~Pooley and P.~King.
\newblock ``{The} unified modeling language and performance engineering''.
\newblock {\em IEEE Software}, 1999.

\bibitem{evol-spn}
A.~Puliafito, M.~Telek, and K.~S. Trivedi.
\newblock ``{The} evolution of stochastic {Petri} nets''.
\newblock In {\em Proc. of World Congress on Systems Simulation}, pages 3--15,
  Singapore, September 1997.

\bibitem{corba-perf}
S.~Ramani, K.~S. Trivedi, and B.~Dasarathy.
\newblock ``{Performance} analysis of the {CORBA} event service using
  stochastic reward nets''.
\newblock In {\em Proc. of the 19th IEEE Symposium on Reliable Distributed
  Systems}, pages 238--247, October 2000.

\bibitem{Schmidt:01o}
R.~E. Schantz and D.~C. Schmidt.
\newblock {Middleware for Distributed Systems: Evolving the Common Structure
  for Network-centric Applications}.
\newblock In J.~Marciniak and G.~Telecki, editors, {\em Encyclopedia of
  Software Engineering}. Wiley \& Sons, New York, 2002.

\bibitem{Schmidt:02a}
D.~C. Schmidt and S.~D. Huston.
\newblock {\em C++ Network Programming, Volume 2: Systematic Reuse with ACE and
  Frameworks}.
\newblock Addison-Wesley, Reading, Massachusetts, 2002.

\bibitem{Schmidt:00a}
D.~C. Schmidt, M.~Stal, H.~Rohnert, and F.~Buschmann.
\newblock {\em {Pattern-Oriented Software Architecture: Patterns for Concurrent
  and Networked Objects, Volume 2}}.
\newblock Wiley \& Sons, New York, 2000.

\bibitem{VEST:03}
J.~A. Stankovic, R.~Zhu, R.~Poornalingam, C.~Lu, Z.~Yu, M.~Humphrey, and
  B.~Ellis.
\newblock {VEST: An Aspect-based Composition Tool for Real-time Systems}.
\newblock In {\em Proceedings of the IEEE Real-time Applications Symposium},
  Washington, DC, May 2003. IEEE.

\bibitem{man-perf}
H.~Sun, X.~Zang, and K.~S. Trivedi.
\newblock ``{A} stochastic reward net model for performance analysis of
  prioritized {DQDB} {MAN}''.
\newblock {\em Computer Communications, Elsevier Science}, 22(9):858--870, June
  1999.

\bibitem{NS-2}
{The VINT Project}.
\newblock {Network Simulator - NS-2}.
\newblock \url{http://www.isi.edu/nsnam/ns}.

\bibitem{williams98}
L.~G. Williams and C.~U. Smith.
\newblock ``{Performance} evaluation of software architectures''.
\newblock In {\em Proc. of the Workshop on Software and Performance}, Santa Fe,
  NM, 1998.

\end{thebibliography}

%% $Id: background.tex,v 1.9 2005/02/15 09:47:44 gokhale Exp $

\section{Background and Related Work}
\label{sec:background}

This section provides an overview of the Stochastic Reward Nets (SRN)
analytical modeling techniques we have used for performance modeling
of DHPS systems. We also describe the related work in this area.

\subsection{Stochastic Reward Nets (SRNs)}
\label{sec:srn_back}

This section describes the background on the Stochastic Reward
Net (SRN) modeling paradigm~\cite{evol-spn} used for performance analysis of
the Reactor pattern. SRNs represent a powerful
modeling technique that is concise in its specification and whose form
is closer to a designer's intuition about what a model should look
like. Since SRN specification is
closer to a designer's intuition of system behavior, it is also easier
to transfer the results obtained from solving the models and interpret
them in terms of the entities that exist in the system being
modeled. 

SRNs are an extension to Petri nets~\cite{peterson81}. 
Petri nets were proposed to represent formally the flow of control in a 
system. A Petri net is a directed graph which comprises two types of
elements, namely, {\em places} and {\em transitions}. A directed arc
connecting a place to a transition is called an {\em input arc}. 
Conversely, an {\em output arc} connects a transition to a place. Arcs
have a positive integer number called {\em multiplicity} associated 
with them, with the default multiplicity associated with an arc being 1.
Places can contain {\em tokens} that move from one place to another
through transitions. A transition is enabled when each of the places
connected to it by its input arcs have at least the number of tokens 
equal to the multiplicity of those arcs. When an enabled transition
fires, a number of tokens equal to the input arc multiplicity is 
removed from each one of the corresponding input places and a 
number of tokens equal to the output arc multiplicity is deposited 
in each one of the corresponding output places. The state of a 
Petri net with $p$ places is represented by a vector $(m_1,m_2,\dots,m_p)$,
where $m_i$ is the number of tokens in place $i$. The state of a 
Petri net is often referred to as its marking. When a Petri net is specified,
it can be made to start at a particular marking called the 
{\em initial marking}. Subsequently, the net evolves by the successive
enabling and firing of transitions which causes the tokens to flow
among places.

Stochastic Petri nets~\cite{evol-spn} extend Petri nets by allowing timed 
transitions that have {\em exponentially distributed} firing times. 
Generalized stochastic Petri nets (GSPNs)~\cite{evol-spn} also allow {\em
immediate} transitions which fire instantaneously. GSPNs include
an {\em inhibitor arc} which can also have a multiplicity 
associated with it. An inhibitor arc inhibits the transition 
it is connected to if the place it is connected to at its other end has a 
number of tokens equal to at least its multiplicity. The default
multiplicity is 1. A GSPN marking with at least one immediate transition
enabled is called a {\em vanishing marking}, and a marking
with no immediate transitions enabled is called a {\em tangible
marking}. 

Stochastic reward nets (SRNs) extend GSPNs further by allowing the association
of a reward rate to each tangible marking. The SRN models allow the concise 
specification of various reward functions. To extend the power of 
specification, SRN includes specification of {\em enabling (or guard)
functions} for each transition. The transition is enabled only if
the enabling function returns ``1.'' SRNs also allow marking dependent
arc multiplicities and enabling functions. Another feature of SRNs
is the provision of priorities and probabilities to determine which
of a set of simultaneously enabled transitions will fire first: the
transition with the highest priority is fired first. If the competing
transitions have the same priority the one to fire first is chosen
probabilistically. 

In a graphical representation of a SRN, a place
is represented as a circle, $n$ tokens in a place are represented by
$n$ dots or the number $n$ within the place, immediate transitions are
represented by thin lines, and exponentially distributed timed transitions
are represented by empty rectangles. An inhibitor arc is represented by a 
circle instead of an arrow at the terminating end. An arc with
multiplicity $m$ is represented by a ``$| m$'' on the arc, and an
arc with a marking dependent multiplicity function is indicated by
a ``N'' or an inverted ``N'' in it. The number of tokens in 
place $p$ is indicated as $p$. 

\subsection{Related Work}
\label{sec:relwork}
Stochastic reward nets have been extensively used for performance,
reliability and performability analysis of a variety of
systems~\cite{corba-perf,avail-vax,poll-perf,man-perf,queue-perf,rel-srn}.
The work closest to the proposed research is reported by Ramani {\em
et al.}~\cite{corba-perf}, where SRNs are used for the performance
analysis of the CORBA event service. UML representations of
application architecture have been mapped to Queuing
networks~\cite{williams98,hoeben00,menasce98,petriu00,pooley99} and
Petri nets~\cite{king99,bonda98}.

Several other analysis tools exist. The Virginia Embedded System
Toolkit (VEST)~\cite{VEST:03} is a model-based embedded system
composition tool that checks whether certain real-time, memory, power,
and cost constraints of DPSS applications are satisfied. The
Cadena~\cite{Hatcliff:03} tool suite provide static analysis,
model-checking, and other light-weight formal methods for component
middleware-based DPSS systems. While these tools are important, they
either deal with other systems features ({\em e.g.}, power
consumption) or have a narrow focus ({\em e.g.}, embedded
systems). Moreover, these tools are applicable once the system
composition decisions are made.

